------

Mauna Loa Solar Observatory Observer's Log

Tue Jul 11 16:19:43 GMT 2000 Year: 00 Doy: 193 Observer: yasukawa WEATHER COMMENT: Tue Jul 11 16:20:08 GMT 2000 Cool, in thick altocumulus and broken cirrus, light southwest wind. Unable to point spar at this time. Tue Jul 11 16:25:35 GMT 2000 CHIP Startup--Initializing new tape COMMENT: Tue Jul 11 16:38:36 GMT 2000 Problem with the slow network were found and rectified. NOAA personnel determined that there were intermittent components in the GONG hub and tranceiver. Between the two devices, the network had slowed to a crawl. NOAA moved us to a working hub and is trying to get GONG to spring for replacing the defective devices. Tue Jul 11 20:00:46 GMT 2000 CHIP ending tape WEATHER COMMENT: Tue Jul 11 20:02:12 GMT 2000 Continued heavy altocumulus overcast, in orographic clouds. COMMENT: Tue Jul 11 20:02:48 GMT 2000 TAPES: MKIV: no data PICS: no data CHIP: no data LOWL: L00731 in drive #1 Tue Jul 11 20:03:50 GMT 2000 MkIV